Author's response to reviews

Title: Memory and Executive Screening (MES): a brief cognitive test for detecting mild cognitive impairment

Authors:

qihao guo (dr.guqihao@126.com)
bin zhou (zhoubin@tri-kobe.org)
qianhua zhao (applenasa@gmail.com)
bei wang (bbllbbll@163.com)
zhen hong (profzhong@sina.com)

Version: 2 Date: 14 March 2012

Author's response to reviews:

The editor sent email to ask me to re-upload manuscript as the editor cannot read their manuscript properly (text cut off at right and left borders).